Author's response to reviews

Title: Evaluation of a system of structured, pro-active care for chronic depression in primary care: a randomised controlled trial.

Authors:

Marta Buszewicz (m.buszewicz@ucl.ac.uk)
Mark Griffin (m.griffin@ucl.ac.uk)
Elaine McMahon (e.mcmahon@ucl.ac.uk)
Jennifer Beecham (JBeecham@lse.ac.uk)
Michael King (m.king@medsch.ucl.ac.uk)

Version: 5 Date: 28 July 2010

Author's response to reviews:

I have made the following further revisions:

Author list: Hopefully all agree now

Affiliations: I've done my best to follow instructions

Tables: Tables 1 and 2 labelled. All lines removed from Table 1 by rewriting it. Table 2 comes up as no lines on my system - please do whatever you want to amend for your formatting

Abstract: please rename the section Results/Discussion to Discussion Done

Figure placement text: please remove this Done

Tables- please remove the visible vertical lines from your tables see above

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space.

I am unable to adjust the figures further - please do what you need to

Files will be uploaded as .rtf and .doc for you to adjust further if needbe